Author's response to reviews

Title: Long term virological, immunological and mortality outcomes in a cohort of HIV-infected female sex workers treated with highly active antiretroviral therapy in Africa

Authors:

Charlotte Huet (charlotte_huet@yahoo.fr)
Abdoulaye Ouedraogo (ouarma74@yahoo.fr)
Issouf Konate (kletio@yahoo.fr)
Isidore Traore (tiandiogo2002@yahoo.fr)
Francois Rouet (franrouet@yahoo.fr)
Antoinette Kabore (aswkabore@yahoo.fr)
Anselme Sanon (anselmes2002@yahoo.fr)
Philippe Mayaud (philippe.mayaud@lshtm.ac.uk)
Philippe Van de Perre (p-van_de_perre@chu-montpellier.fr)
Nicolas Nagot (n-nagot@chu-montpellier.fr)

Version: 4 Date: 31 August 2011

Author's response to reviews:

This is the final version of the paper, as you requested, having done only some formatting edits (removed highlights) - in a couple of places the word 'participants' has replaced the word 'patients', thus not altering word count.